Author's response to reviews

Title: Effect of roflumilast on inflammatory cells in the lungs of cigarette smoke-exposed mice

Authors:

Piero A Martorana (martorana@unisi.it)
Benedetta Lunghi (lunghi@unisi.it)
Monica Lucattelli (lucattelli@unisi.it)
Giovanna De Cunto (giovanna.dec@libero.it)
Rolf Beume (rolf.beume@nycomed.com)
Giuseppe Lungarella (lungarella@unisi.it)

Version: 4 Date: 14 August 2008

Author's response to reviews:

1) An acknowledgements section has been included accordingly:

Acknowledgements
"This work was supported with funds from the University of Siena, Siena Italy (PAR grants to ML and GL). Nycomed GmbH funded in part some of the expenses incurred in the course of this study and covered the publication expenses."

2) The title has been modified according to the suggestions:
"Effect of roflumilast on inflammatory cells in the lungs of cigarette smoke-exposed mice"